

MDPI

Article

Social Sustainability Work in Product Development Organizations: An Empirical Study of Three Sweden-Based Companies

Patricia Lagun Mesquita * and Merlina Missimer

Department of Strategic Sustainable Development, Blekinge Institute of Technology, SE-37179 Karlskrona, Sweden; merlina.missimer@bth.se

* Correspondence: patricia.lagun.mesquita@bth.se

Abstract: Companies responsible for product development (PD) and manufacturing play an important role in supporting society in achieving sustainability, and yet they do not always include full sustainability considerations in PD decisions. The social dimension of sustainability has been largely neglected in the PD field and there is very little empirical evidence of social sustainability implementation in general. The aim of this paper is to investigate how social sustainability is currently included in PD organizations and what their motivations are to do so. Results from a comparative case study approach with three organizations include rich descriptions across four dimensions: The scope of the work and their view of interdependencies with other social systems, their definition of social sustainability and the issues they work with, what guides strategic decisions, and how this internal work is structured. The results reveal that the three product development organizations are heterogenous in their approaches to social sustainability and that the more advanced approach shows a better understanding of the complexity of social sustainability and a broader perspective of its interdependencies, which goes hand-in-hand with a way of organizing that overcomes traditional hierarchies and allows for more collaborative and strategic work in this area. This systems perspective also drives what issues are included in an organization's work; scope and definition of social sustainability become more encompassing and aligned. Finally, our study shows that social sustainability impacts connected to products' lifecycles, when addressed, are done so by functions outside design activities, as opposed to product developers. A greater understanding of how companies currently approach social sustainability and what challenges they might face in integrating it in organizational and design related practices has been called for; our paper contributes to this but acknowledges that more work is needed.

Keywords: social sustainability; product development; strategic sustainable development; case study



Citation: Mesquita, P.L.; Missimer, M. Social Sustainability Work in Product Development Organizations: An Empirical Study of Three Sweden-Based Companies. Sustainability 2021, 13, 1986. https://doi.org/10.3390/su13041986

Academic Editors: Glenn Johansson and Zenon Foltynowicz Received: 15 January 2021 Accepted: 10 February 2021 Published: 12 February 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/by/4.0/).

1. Introduction

1.1. Sustainability in Product Development

Worldwide production and consumption of products are a key cause of ecological and social problems that we face today [1,2]. Sustainability impacts can occur at many stages of a product's life-cycle—from raw material extraction and acquisition, production, distribution, use and maintenance, to end-of-life reuse, recycling, treatment or disposal, and are known to be largely determined by decisions taken during early stages of product innovation, that is, during product development [3]. Companies that are responsible for product development (PD) processes and manufacturing, therefore, can play an important role in supporting society to achieve sustainability and can, if doing so strategically, gain business economic benefits. In this study we refer to these companies as product development organizations (PDOs).

Starting from the 1960's, attempts to find solutions to environmental problems were mostly reactive, focusing on "end-of-pipe" and "repair" solutions [4,5]. After major acci-

Sustainability **2021**, 13, 1986 2 of 20

dents and disasters, such as the Bhopal gas tragedy and the Sandoz chemical spill in the 1980's, public pressure led industry to take more responsibility for its actions. According to Simons et al. [6], two other generations of more proactive environmental strategies unfolded in the subsequent decades. Starting with pollution prevention and eco-efficiency strategies and moving on to life-cycle management and lifecycle thinking, the focus turned towards avoiding problems before they arise. Companies began to consider measuring the ecological aspects of products over their entire lifecycles and to attempt to use this knowledge to improve the products. This shift gave rise to a new wave of studies focusing on integrating sustainability considerations in the PD process. As part of these efforts, several concepts, methods and tools were developed, such as eco-design [7], design for environment (DfE) [8], and Life-Cycle Assessment (LCA) [9]. Amounting to over 600 unique tools and methods, these have in recent studies [10,11] still been deemed insufficient when it comes to facilitating sustainability integration in existing company processes and a resulting adoption by industry as a whole.

1.2. The Social in "Sustainability"

Although there is widespread consensus of the concept of sustainability being composed by at least an ecological and a social dimension, the above-mentioned tools have largely focused on ecological aspects [12–14]. Gmelin and Seuring [14], when looking at the determinants of sustainable new PD, stress how the social dimension of sustainability has been largely neglected. This neglect of the social dimension in sustainable product development (SPD) research mirrors the general sustainable development field [15–20]. Due to the lack of agreement of how to define it [19,21], social sustainability therefore ends up as a "laundry list" [22], or "black bin" [23], where all the indefinable elements of sustainability seem to be placed.

However, for PD to be considered sustainable, the social aspects must also be taken into account, and support for dealing with social challenges require further research and development [14] (p. 2). An initial literature search combining the terms "social sustainability" with "product design", "product development" and "manufacturing" led to few results and little insight into how to integrate social sustainability in product design and development or how PDOs systematically work with social issues. On a product level, the greatest development has been on social lifecycle assessment methods, which up to date still lack a unifying framework that could allow the further development of the field and turn Social LCA into a more robust decision support tool for PDOs [24,25]. Social LCA, like many other tools and methods related to SPD, tend to be overly complex, focus too much on technical assessments, and organizational complexities are not understood enough to allow for successful integration of sustainability considerations in business operations. [10,26,27]. Another great challenge is that without a clear understanding of what social sustainability in fact means, it is hard for companies to assess how they and their products contribute or not to a sustainable society. Current methods and tools thus provide little insight in terms of supporting PDOs in strategically integrating a social sustainability perspective.

1.3. Empirical Evidence of Social Sustainability Integration in PD Organizations

There seem to be little literature using empirical data to follow up on the implementation of social sustainability or on the organization of social sustainability work in general. In preparation for this project, we conducted a literature review using google scholar, the scopus and web of science databases as well as a database including PhD dissertations (summon.bth.se). The aim was to find research that included empirical evidence of how companies worked with social sustainability. So not what they are supposed to do (i.e., theoretical propositions of tools and approaches), but what they actually do and how they organize their work. The search focused on the terms "social sustainability" in combination with the terms "business" OR "corporate" and "management processes", "implementation", "organizational processes", "organizational structures", "management practice" OR "governance" in various combinations of title, abstract and keyword search. This search

Sustainability **2021**, 13, 1986 3 of 20

produced more than 200 results, including duplicate results. Yet, on closer investigation, very few articles and sources included empirical data on social sustainability work in companies, especially PD ones.

Schulte and Hallstedt [28], who created a self-assessment method of the integration of sustainability in PD combining several maturity models and specifically added the social sustainability dimension in their questionnaire, show that the participating companies struggle with integrating social sustainability all the way into PD. The study, however, does not explain why it is so or what efforts currently exist to address the social dimension. Watz and Hallstedt [29], when conducting a multi-case study with seven PDOs, found that, in terms of integration of sustainability requirements in the PD process, only one organization had begun considering social aspects alongside ecological ones. Out of their five-level scale of sustainability requirements integration, that same company featured on level four, showing that even organizations with high levels of maturity in integrating ecological sustainability in PD struggle with doing the same for social sustainability.

Wang et al. [30] in their editorial to the Academy of Management Review call for a descriptive research agenda that investigates *how* and *why* corporations actually engage in social sustainability (framed as Corporate Social Responsibility) to only then derive a better understanding of *how they should* engage in social sustainability. The aim of this paper is to do exactly this, to provide empirical evidence of how social sustainability is currently included in PDOs and what their motivations are to do so. In meeting this goal, we also seek to begin addressing a more normative need, to support organizations to strategically contribute to overall social sustainability. Our research question is: *How do product development organizations strategically work with social sustainability?*

The article is structured in five parts. Section 2 summarizes the research approach and the different methods used. Section 3 provides in depth data on social sustainability integration in three case study organizations and Section 4 provides discussion and conclusions, and Section 5 discusses limitations.

2. Research Design and Methods

2.1. Comparative Case Study Approach

Case study research has been defined as "a strategy for doing research which involves an empirical investigation of a particular contemporary phenomenon within its real-life context using multiple sources of evidence" [31] (p. 146). In this paper we use a comparative case research approach, which still seeks "thick descriptions" common in single case studies, but the main goal is to discover contrasts, similarities, or patterns across the cases with the intention to contribute to the development or the confirmation of theory. By choosing this approach we were able to go beyond the specificities of a single case, and through careful abstraction [32] provide a coherent framework to answer our research questions.

Analysis in the comparative case approach is iterative. This led to a sequential selection of cases (company A, then B, then C) and an initial analysis of one case had to be conducted prior to the selection of the following case. Organizations were then purposefully sampled from a group of companies engaged in an existing project based on their likelihood to present data-rich cases [33] on social sustainability integration.

2.2. Data Collection

In total, we collected data from three multinational PD and manufacturing organizations based in Sweden. A total of 21 semi-structured interviews of roughly 60 min were conducted. In organization A, a large product development and manufacturing company (about 2000 employees), we conducted 13 interviews between the years 2016 and 2017. In organization B, a large construction machine manufacturer (4000 employees), we conducted 4 interviews in the year of 2017. In organization C, a large construction machinery manufacturer (about 1200 employees), we conducted 4 interviews in the year of 2019. Selection of interviewees was done in dialogue with the organizations and in accordance with research needs and therefore interviewed roles varied see Table 1.

Sustainability **2021**, 13, 1986 4 of 20

Table 1. Organizations and interviewee roles.

Organization	Function		
	Safety and occupational health		
	Research and development/product sustainability		
	Purchasing		
	Environmental management		
A	Communication		
	Manufacturing process owners for four different processes		
	Product support and after market		
	Quality		
	Compliance		
	Design leader		
	Sustainability and Corporate Social Responsibility (CSR)		
_	Marketing and after sales		
В	Research and development		
	Human resources		
	Purchasing		
	Product portfolio		
С	Human resources		
	Environmental health and safety		

In organization A, for example, we started with interviews at the operational level of product development activities to quickly realize that social sustainability concerns were foreign concepts to the interviewees and that we needed to search for insights elsewhere (also to understand why that was the case). Together with the research project leaders inside organization A we then selected a range of other roles at the management level, some directly connected to PD activities and others in adjacent areas such as purchasing, environmental health and safety and communication. When selecting the interviewees in organizations B and C we requested similar roles to take part in the project for the sake of comparison.

In the first part of the interview, participants were asked to provide their definition and understanding of social sustainability, describe how their organization defined and implemented goals related to social sustainability and explore what it meant for their individual role. We then provided the interviewees with the definition of social sustainability used in this study (see Section 2.3.1) and asked if they would add any projects, initiatives or practices given this new understanding.

Interviewees from organization A more directly involved in product development activities were also provided with an environmental impact assessment (EIA) of one of their products and asked to describe to the best of their knowledge the social implications of different lifecycle activities and how actively managed or not those aspects were. The purpose with the exercise was to create a common understanding and language of lifecycle activities and potential social sustainability problems related to these, to then explore in what ways the organization was or could be handling those problems. As the exercise yielded limited insights due to a lack of awareness of social sustainability problems, we adjusted the interview protocol for organizations B and C to cover broader organizational and decision-making aspects.

A workshop was also conducted with participants from organization A and B. The workshop was part of a larger project and company C had not yet been selected at the time

Sustainability **2021**, 13, 1986 5 of 20

of the workshop. A force field exercise was conducted to explore barriers and incentives in the integration of social sustainability in their organizations.

All interviews were tape recorded and transcribed verbatim. Data from the workshop were collected in the form of observation during the session and the force field analysis produced by the participating organizations on a flip-chart sheet. To ensure data reliability, we triangulated the interview statements with publicly available as well as internal company documents (e.g., sustainability reports, codes of conduct, training manuals, etc.).

2.3. Data Analysis

Data from the interview transcripts were coded in multiple rounds using thematic analysis [34]. Data from case company A were coded inductively; the emergent themes were then used in the coding of data from case company B and C. We created subcategories when the new data fit into existing categories but were more specific in nature. In a new round of coding, we used the five-level model within the Framework for Strategic Sustainable Development [35], and specifically the social dimension of its definition of sustainability [36,37], to organize the data and interpret the findings. The interview data and the data obtained from the workshop and the company documents were triangulated, revealing overall consistency in the findings.

2.3.1. The Framework for Strategic Sustainable Development

The Framework for Strategic Sustainable Development (FSSD) is intended to provide an overarching and unifying framework that is based on a systems science understanding and that allows for strategic planning as well as strategic selection and design of tools to help move any organization towards sustainability. It includes a five-level structuring model that suggests that, for strategic planning, it is of importance to agree upon the system (level 1) that is to be planned within, and only then to go on to define success (level 2) within that system. Once success is defined, strategic guidelines (level 3) can be used for the selection and prioritization of actions (level 4); all four of these levels can be supported with various concepts, methods, and tools or other forms of support (level 5) [35,38,39]. It is the interplay between defining success and clarifying the system boundaries with that in mind, that leads to the breakthrough in strategic thinking and planning. Table 2 describes the five levels in the context of strategic sustainable development.

Table 2. Five-Level Model of the Framework for Strategic Sustainable Development.

The System Level	describes the overall major functioning of the system, in the case of sustainability the organization within the human society within the biosphere.	
The Success Level	includes the definition of the vision, bounded by basic sustainability principles.	
The Strategic Guidelines Level	specifies guidelines for how to approach the principle-framed vision strategically, in an economically viable step-by-step approach.	
The Actions Level	comprises everything done in concrete terms, the actions that were prioritized into a strategic plan.	
The Tools Level	includes concepts, methods, and tools and other forms of support that are often required for decision support, monitoring, and disclosures of the actions to ensure they are chosen in line with the strategic guidelines to arrive step-by-step at the defined success in the system.	

The sustainability principles of the FSSD (three of them being about the ecological system and five of them being about the social system) are designed for being useful for "backcasting" from a principle-framed vision, acting as boundary conditions for sustainable solutions. Backcasting means imagining success in the future and then looking back to

Sustainability **2021**, 13, 1986 6 of 20

today to assess the present situation through the lens of this success definition, and to explore ways to reach that success [40]. The social sustainability dimension specifically is built on a complex adaptive systems understanding of human social systems and the mechanisms that destroy this social web. By re-framing the mechanisms of destruction as design constraints and declaring these as success or the goal, the approach can support organizations at all levels to strategically contribute to sustainability. Since we are interested in the potential contributions of products and PDOs to social sustainability, we here present only the principles related to the social system that were also used to clarify the meaning of social sustainability during the interviews. The Social Sustainability Principles (SSPs) of the FSSD posit that, in a socially sustainable society, people are not subject to structural obstacles to (1) health, (2) influence, (3) competence, (4) impartiality, and (5) meaningmaking. Structural obstacles here refer to social constructions—political, economic and cultural—which are firmly established in society and upheld by those with power. Due to various kinds of dependencies and other factors, such obstacles are difficult or impossible to overcome or avoid for those affected by them [37].

While the five-level model was useful to structure the data and create a frame for interpreting how the case organizations, in their own terms, approach social sustainability, the whole FSSD was key to assess how their current approaches strategically contribute to (socially) sustainable development. The FSSD brings some advantages that lend themselves favorably as a basis for strategic planning and assessment. Besides providing a principled definition of social sustainability and thus responding to some of the challenges raised in Section 1.2, it is deemed useful as [41]:

- It is built on a scientific systems-understanding of both the natural and the social system.
- It includes a clear definition of sustainability, which due to its principled nature is generally applicable and yet concrete enough to guide action.
- It facilitates stepwise and strategic movement towards the goal of sustainability.
- It helps make better use of other tools, methods or frameworks for sustainability: once it has been used to identify the gap to sustainability, as well as desirable strategies to bridge the gap, it is easier to select and inform other forms of support needed for the transition [39].

3. PDOs and Their Approach to Social Sustainability

In-depth interviews with 21 professionals of three different large PDOs provided several insights into current organizational practices related to social sustainability and also into barriers and opportunities for PDOs to better contribute to social sustainability. Having interviewees in different functions such as manufacturing process leaders, project leaders in design and development processes, purchasing, R&D, health and safety, quality, human resources, communication, marketing and after sales, environment and Corporate Social Responsibility (CSR), and at operational, tactical and strategic levels of the organizations, allowed us to identify the locus of different aspects of social sustainability work and how integrated they actually are into practices of the organizations investigated.

Our findings are presented in four dimensions that characterize an approach to social sustainability and also provide the ability to distinguish essential attributes to each approach. The first three dimensions follow the five-level model for strategic planning in complex systems used in the FSSD. These dimensions are interrelated and help to inform each other. The fourth dimension, *how internal work is structured*, was added to support characterizing the realization of the approaches beyond static statements that can be found in company sustainability performance reports. Examples of actions and tools (levels four and five of the FSSD) were also identified and were used to illustrate enactment of the previous levels. Below we provide a brief description of the dimensions:

System boundaries: this dimension is about how organizations define the scope of their social sustainability work, both internally to the organization (stakeholders, roles, departments and areas of the organization that should be involved in the work) and

Sustainability **2021**, 13, 1986 7 of 20

externally (other organizations, stakeholders, value chain actors). In this dimension organizations display their recognized interdependencies with other social systems, their sense of responsibility and their perception of what can be influenced and/or controlled.

Success: in this dimension organizations display their definition of social sustainability, what the common concerns and issues are, what it means to be successful in this regard, and in what ways such a definition is related to the organization's own success.

What guides strategic decisions: this dimension is about the organization's drivers and motivations for including social sustainability concerns in the planning and development of organizational action, and how priorities are defined.

How internal work is structured: this dimension displays the organization's choices for structuring social sustainability work including how goals and priorities are operationalized throughout the organization.

In Table 3 we summarize the key findings in each of the dimensions and for all three organizations.

Table 3. Summary of findings according to the dimensions of the companies' social sustainability approach.

Dimensions	PD Organization A	PD Organization B	PD Organization C
System boundaries	 Narrow view of interdependencies with other social systems beyond own organization Focus on own employees' wellbeing and their immediate surrounding community Product lifecycle aspects limited to safety concerns at own manufacturing and use phase First tier suppliers expected to comply with company's code of conduct (no evidence of verification) 	 Broad view of interdependencies with other social systems Engagement with actors upand downstream the value chain Product Lifecycle thinking driving how system boundaries are set 	 Narrow view of interdependencies with other social systems beyond own organization Focus on own employees' health and safety Product lifecycle aspects limited to safety concerns at own manufacturing and inconsistently at use phase First tier suppliers expected to comply with social sustainability related requirements (ad hoc verification)
Success	 Weak knowledge and awareness of social sustainability concepts Social sustainability agenda defined by expectations from local/national stakeholders and industry requirements Existing goals for employees are defined at site level and for product social performance by industry standards Organization's success not seen as dependent on thriving social systems 	 Good knowledge and awareness of social sustainability concepts Social sustainability agenda defined by international standards and multi-stakeholder expectations Goals are defined at corporate level and embedded in the corporate culture Organization's success seen as dependent on thriving social systems 	 Uneven knowledge and awareness of social sustainability concepts Social sustainability agenda defined by demands from local/national authorities Existing goals for employees are defined at site level and for product social performance by customers Organization's success not seen as dependent on thriving social systems
What guides strategic decisions	 Compliance and competitiveness Risk and cost avoidance Present and future customer needs 	 Branding and reputation through core values Building trust among societal actors 	 Branding through product quality Compliance and competitiveness Risk and cost avoidance Expressed customer needs

Sustainability **2021**, 13, 1986 8 of 20

Table 3. Cont.

Dimensions	PD Organization A	PD Organization B	PD Organization C
How internal work is structured	 Code of conduct defined at top management level but no clear processes to "bring it to life" Social sustainability aspects not explicitly anchored at any level of the organization Uncoordinated efforts at the tactical level and no structured collaboration between functions First initiatives at product development level but without clear possibilities to influence higher levels of decision-making 	 Social sustainability anchored alongside ecological sustainability at the strategic level of the organization, where strategic guidelines are defined Multiple managerial functions develop specific strategies aligned with corporate values Vertical flow of information and support for decision-making Horizontal collaboration through cross-function dialogue routines 	 No social sustainability guidance from top management Social sustainability aspects not explicitly anchored at any level of the organization Uncoordinated efforts at the tactical level and no structured collaboration between functions

The following detailed account of the product development organizations (PDOs') approach to social sustainability further explores the findings presented above.

3.1. System Boundaries

Interviews indicate that the system in focus at company A is the organization itself and the relationships between the people directly employed by it and their immediate surroundings, such as employees' families and the communities that they are part of. When asked about examples of projects, initiatives or guidelines related to social sustainability, multiple interviewees refer primarily to the strong safety culture of the organization, from factory floor to office environment. Alongside the safety campaigns and procedures, interviewees were quick to mention efforts of being part of the local community by sponsoring sports associations and local clubs. A recent change has been to expand the work on safety to include broader employee health, so to promoting work-life balance, reduce stress and help prevent burnout. The occupational health representatives noted, though, that other subsidiaries of the organization outside Sweden performed much lower in this area to the point that it would be considered non-compliant from a Swedish perspective: "there the employees work in their own clothes. They haven't got jackets. They wear their own private clothes and a lot of dangerous things, chemicals on the shop floor. (. . .) Here we see the whole employee, both soft issues, social at home, the work you do." Organization A thus seems to determine the scope of its social sustainability work based on local and national contexts and expectations.

In terms of their products' lifecycle, only some aspects of lifecycle activities or phases are included in their scope of social sustainability related activities. Interviewees in functions connected to product development recognized the existence of potential social impacts in their supply chain but explained that currently only first tier suppliers are expected to act in accordance with their code of conduct, and it is unclear how compliance with the code of conduct is guaranteed. Interview participants involved in purchasing related questions reported to usually rely on industry level supplier certification processes and did not know if requirements specifically related to social sustainability were part of certification and verification processes. The guiding document for supplier code of conduct, binding by contract, also does not guarantee compliance, and basically relies on supplier responsibility by asking them to contact the company in case they think they are not complying with such code of conduct. Social impacts connected to raw material extraction were mentioned by product developers as problems that should be addressed by product design or purchasing procedures at the same time that they recognize the lack of tools and organizational man-

Sustainability **2021**, 13, 1986 9 of 20

date to allow necessary changes to happen. During the product use phase, safety is the one concern that interviewees can connect to social sustainability, as highlighted by one interviewee: "the way we do our work here, we need to have this product safety very high in all our systems". Such requirement, however, is externally demanded by the industry regulations and certification needs.

Interviews with organization B representatives, in contrast, indicate that the company includes both upstream and downstream value chains in the scope of its social sustainability work. Moreover, it is through considering their products' lifecycle that value chain actors are identified and engaged in developing solutions to social sustainability challenges: "what we really (. . . are) trying to work with now, is this responsibility in the broader value chain, or in the whole value chain. And in that case, you start of course, with product development. You can start with a supplier. You go into your own operations and the next step is the sales and our customers or market. And so social responsibility is that we make sure that we act responsibly in all those different value chain parts. And in terms of suppliers-where I worked a long time in different kind of assessments and so on-in terms of supplier responsibility. Making sure that our suppliers are not dealing with child labor and that kind of things. No corruption and these kinds of topics. In our own operations, of course, we have our own code of conduct which is on the wall here. Which states very much how we should act both in the other environment, but also in the social dimension. And the same goes there, of course, with child labor, corruption, that we're very much act according to the human rights and so".

Employee wellbeing is seen as deeply embedded in the corporate culture and also the starting point of the evolution of the social sustainability work in the organization. This corporate culture also travels well to all of the organization's subsidiaries. One interviewee mentioned that entering another site in another country feels like going into their own Sweden-based site. Standards are the same, values are the same and people treat each other the same way. Legal standards play a role in the sense that they need to be followed, but the organization claims to have a performance expectation that is above legal standards and that are applied to all of its operations.

Interdependencies with other social systems beyond the own organization in different lifecycle stages are recognized and different initiatives are developed to create positive social impacts. Interviewees mention such initiative as win—win situations, where business can prosper as a result of well-developed societies and trusted relationships: "I think vocational schools is a very good example, which we have—we are now running three vocational schools in Africa. We're running one in China and probably one or two others, both developing the country and society but, of course, also our business. So both sides."

From a product lifecycle perspective, social sustainability requirements are placed both on suppliers as on customers (dealers), who have recently started to use a screening process before contracts are signed. The screening includes similar aspects from the supplier code of conduct, as one way to avoid the products to be used in situations that may cause harm to people—"What we're doing the most now we're putting a lot of efforts is to make a similar kind of assessment, or screening of our higher risk relations, in terms of sales and marketing. How our products are used in the market? So they are used in a responsible way but also the customers that we're selling our products to. That they are acting responsibly and they have similar kind of requirements as we have in our own code of conduct. Again, getting back into the same kind of questions with corruptions and these topics as you mentioned; child labor, human rights, etc."

Interviewees also speak about how the organization works systematically with suppliers and supply chain traceability in order to influence aspects that they may not have direct control over, such as second and third tier suppliers. Although this illustrates how the organization operationalizes strategic social sustainability goals, it also shows that the scope of what is considered as the organization's responsibility is considerably wider than in organization A. Interviews also revealed that social sustainability related requirements are often defined at higher levels of decision-making and work more as design constraints that are set outside product development activities per se. Exceptions are

Sustainability **2021**, 13, 1986 10 of 20

product safety and safety of manufacturing processes, which are very much influenced by different design alternatives.

Organization C, similar to organization A, places its focus on the functioning of the organization itself and the internal relationships that are needed for the business to thrive in its own local context. Human resources, for example, when asked to describe the scope of the work done in the area explains it as "hiring and firing. That is, of course, the basic. But one important thing is that we all follow the collective agreements and the law fulfillments and all the things around that, of course, taking part of my time." Unions in the Swedish context play a vital role in shaping the organization's HR scope and employee related programs are a direct result of demands placed by active unions. Legal compliance appears to guide what human resources prioritize in their agenda, and since regulatory frameworks vary from country to country, it is unlikely that organization C uniformly sets a social sustainability scope and performance standard. Engagement or projects with the local community were not mentioned in the interviews.

Local and national authorities play an important role in setting the scope and the standard to be achieved when it comes to occupational health and safety and, similar to organization A, when operating in countries with lower regulatory requirements, organizational practices will also at best meet the local requirements. One interviewee says: "We are very quick here (in Sweden) to report incidents, accidents. But if we go to India, there are more or less no accidents at all, absolutely no incidents. That's a huge difference. The same in China." The interviewee then explains that this difference is probably due to a different cultural understanding of what an incident or accident is and what is worth reporting. Employees might hurt themselves but not miss work, which in turn is not reported. In addition, although subsidiaries and first tier suppliers are recommended to use certifications like OHSAS 18001 or ISO 45001, they do not necessarily do so, and non-conformities are often found during occasional site visits. "It depends also or related to the local laws. Here we need to - we have to follow the laws here. And then we have to fulfill it. And, of course, we want to be a little bit better. But I think if we have this level of law, since they have this level of laws in India (showing different heights with the hands), they will fulfill the laws, but not more."

Another interviewee, whose role mainly involves purchasing decisions at product development, adds that "with new suppliers, it needs to be audited. That's the basic rule. Reauditing is mainly linked to problems that need to be solved, etc. That's when we reaudit a supplier. It's not like any structured plan. We don't have the resource to do that." When asked about how problems would be identified then, the interviewee answered: "I mean, that can be linked to quality problems or whatever, not linked to the social sustainability or anything like that. Those things we will not find unless we make an audit, and maybe not even then because it can be really hard to find those." The supplier checklist used when signing new contracts and also on the event of a site visit, covers a wide range of social sustainability concerns, such as child labor and forced labor, work environment, health and safety, discrimination, freedom of association and more, indicating that at some point it was decided a supplier's social sustainability performance was relevant to organization C's own social sustainability scope. However, apart from limited resources to perform regular audits and the challenges in verifying potential misconduct, the interviewee also mentions the lack of employee training on these issues to allow more thorough supplier site visits. So, with organizational resources being scarce, and problems being hard to track, the general responses become a reactive one: deal with problems in the case that they show up.

In terms of their products' lifecycle, in addition to supplier concerns, company C's social sustainability efforts also include health and safety aspects related to product use. One interviewee refers to their compacting machines as a work environment in itself, therefore the wellbeing of machine operators is considered an important driver in product design. One interviewee says that this is true however only for markets that also share values related to safe work environments and are willing to pay more for extra features. "It's a complex market situation since we're working on a global scale. We have everything from machines that look like they did 50 years ago to what they preferably should be in 5 years' time."

Sustainability **2021**, 13, 1986 11 of 20

Within product design, issues related to conflict minerals were only once brought up by top level management when the company still belonged to the previous group and according to the respondent, at the time they deemed that the issue was not relevant to their product portfolio and even if it was, it would be very difficult to trace of influence changes so far up in their value chain. Unless it were to become a matter of legal compliance: "If we would do something, I mean, to try to further investigate or to go further down the supply chain or up—oh, yeah, down, I can't see that happening, actually, unless there is some specific action required like by law or something like that."

3.2. Success

When asked to reflect what social sustainability meant in the context of their business, none of the interviewees from organization A were able provide a company's definition that encompassed different themes usually placed under the concept. Several "scratched their heads", seemed confused, and asked for further clarification. After we gave them our definition of social sustainability, some were able to immediately relate to the work on safety in manufacturing processes and also the importance of creating a healthy work environment where people manage their workload well and feel that they contribute to the success of the organization in a fulfilling way. A few interviewees in fact expressed concerns that the organization did not seem to have a strong vision any longer to contribute to a purpose beyond economic ones. They explained that the organization used to be part of a bigger group that had clear core values and that those transpired throughout the organization. The health and safety culture remained after the change of management, but the higher purpose was lost and the organization turned inward, focusing on costs and avoiding risks. Themes typically included in a CSR agenda like human rights violation, discrimination and fair business practices, although featured in company's policy, supplier code of conduct, marketing material, and company vision, were not listed as social sustainability concerns, translated into concrete organizational practices, and not related to any of the interviewees' roles in their own perspective. These issues were considered important and relevant from a societal perspective, but not a responsibility of the organization besides taking care of their own employees. The interviewees directly connected to our research project, not surprisingly, expressed their wish to integrate social sustainability aspects (from a value chain perspective) in their product development practices despite the perception of lack of top management commitment to the topic. However, they felt that it was unclear what could actually be achieved considering the perceived lack of interest and direct communication channels with higher levels of the organization.

Success for organization B appears to go together with achieving social sustainability on a societal level. "When we talk about corporate social responsibility, then we talk about our own part into it (sustainability). And for us, it means that we should enact with business ethics, in a good way. We should live up to human rights, and make sure that we treat people fairly." One interviewee also mentions the importance of the individuals in the organization connecting with the organization's purpose: "To me, it's very important that my team, my organization, we as a team, make more than just money. (. . .) We are involved in developing society. But by being successful as a business compass, we can actually drive the society forward." At least, that is the vision communicated by top management and routinely embedded in the culture of the organization. All four interviewees at company B easily related to topics commonly included in the social sustainability agenda and provided similar accounts with respect to core values, that are lived throughout the organization and reflected in the company's practices. Another interviewee speaks to living up to this vision consistently throughout the organization: "Well, when it comes to business ethics, it could also be—even if we follow the law, for example, it could not be enough for us. We, as a company, want to conduct business in a certain way because we think that's ethical. So we can even have much stronger requirements on our sales than what the law require us to do. And in certain cases, we might think that there are too big differences between different countries in terms to the local laws, regulations, and we might choose a

Sustainability **2021**, 13, 1986 12 of 20

higher practice. So our standards might be a little bit higher than what that local market recommends or what the law says in that market. So I think that's where business ethics comes in." Interestingly, all the interviewees were well-versed in the social sustainability challenge and could easily describe how related objectives connected to their own roles in the organization, indicating the presence of a good level of coordination between the overall definition of social sustainability at the top of the organization and the implications for different areas of the organization.

Interviewees at organization C were not as confused about why they were being interviewed as in organization A but we found that the level of awareness of social sustainability as a part of the sustainability concept was very uneven. One interviewee in a purchasing function, for example, could vividly describe how he sees the purpose of organization C in the context of social sustainability: I like to be a bit grand when I talk to our salespeople because we build a foundation of the whole society in the sense that everything we do—most of us, every day—relies on proper compaction. If this ground wasn't compacted properly, the house wouldn't have been standing here for 50 years. If the roads are not built in the right way, transportation won't work, communication. We're talking about digitalization and connected societies. But in order to have connectivity, you need to build (...) 5G communication. They have to have a stable foundation that somebody has to compact. So if you go that far, our greatest impact on social sustainability or sustainability for the society is that the machines perform the way they're supposed to, and that we can help educate and support the users so that they do the best job." Although this vision speaks to the social impacts of the organization's product offerings during its use phase, it fails to include social sustainability impacts that are created in the remaining lifecycle stages, almost as if the ends justified the means. Other interviewees in organization C did not provide any specific vision of success regarding social sustainability but expressed a need of doing more in this direction.

3.3. What Guides Strategic Decisions

When it comes to guiding strategic decisions, several interviewees in organization A point out that the main driving forces are compliance and competitiveness, not goodwill. Representatives of the company during the workshop have also acknowledged several obstacles for integrating social sustainability into company strategy, such as cost and efficiency pressure, the lack of examples from the industrial sector that show the benefits of a social strategy and that they could follow, and lack of internal prioritization of social issues. When it comes to tracking and measuring social aspects in the supply chain workshop participants from organization A have also raised the problem of costly and lengthy assessments and the lack of internal awareness and specific knowledge to make use of assessment tools and results in a meaningful way. Interviews also showed that there is a strong message coming from higher levels of decision-making that cost is a critical variable and that for (social) sustainability to become a priority it needs to be monetized. The interviewees directly involved in this research project express this to be a challenge, as currently there is not enough knowledge or tools that can help justify additional efforts in social sustainability in monetary terms. Customer needs are the key driver for this organization that has other bigger industrial players as end customers. These customers usually operate within fairly long time horizons, which requires company A to include industry trends and potential future customer needs. Moreover, being part of an industry that is highly regulated and dictates product specifications from the outside-in, freedom is perceived to be limited. Since social sustainability specifications are not on the industry's agenda, interviewees in company A do not see a compelling business argument for more social sustainability integration. Relationships with close stakeholders such as the surrounding community, authorities and industrial customers do not flag the need for further action and, as long as these relationships are well managed, the organization will not change its approach.

Branding and reputation are identified from the interviews as key drivers for organization B when making decisions about social sustainability challenges. Social risk

Sustainability **2021**, 13, 1986 13 of 20

management falls under the work referred to as "core value management" that focuses on infusing the organization's core values in decisions taken at all levels of the organization. "It becomes very important when you talk about reputational risk for a global company because we are not local in one market, we are global. So if you do something in one market, it will have echoes into other markets, and it will impact our possibility to do business worldwide. So it becomes very important for us to have policies, guidelines, and support for the business responsible to conduct business in as ethical way as we can. Of course, it won't be perfect, but at least they're setting up some guidance and some support for the business owners to be able to do this as a stringent and equal all over the world as possibly even as the challenges are very different in different markets."

Trust and being a trusted partner are highlighted as one of the values that guides the work: "Business is made by people. Business starts with relationships so of course, the more stable the communities, the societies, the easier it is to conduct business in an ethical way and the better relations we have with a society, the more positive society will be towards us as a brand also. So a lot of these values are built in". Furthermore internally to the organization: "another area that is very important, or one of the values, the trust. This is about the respect for the individual and that we are transparent. (. . .) that we are empowered and that we have the courage, right. Collaboration is important. And that we as managers or leadership trust that our employees do their best, and that we actually can expect them to make decisions and to trust that they do take decisions that is the best for the company."

Connected to branding and reputation, one interviewee raised the need for society in general to expect more of companies in order to accelerate social sustainability improvements: "we are an engineering company and (. . .) we are quite good in following norms and requirements and these kind of things. So I mean to really take the next step, I think the society needs to challenge us to take the next step and put up some boundary conditions. And then we are quite good to adapt to that." If expectations are higher, companies that profit from such social capital will want to do more. However, this is not perceived as a reality yet in company B's context. Talking about customer-driven requirements one interviewee said: "I wouldn't say that these kind of requirements (social sustainability requirements) are that frequent. I would say it's a big discussion when there is a big site, most likely governmentally financed. Rebuilding trains or roads or big infrastructure investments that the government during their process are writing requirements on fuel consumption, driving CO₂ emissions down, etc. More focusing on the environmental side."

Interviews in organization C revealed some of the existing drivers for integrating social sustainability in the organizations. One interviewee in the human resources function talked about the importance of a certain leadership style that favors positive outcomes on social sustainability and referred to the "Swedish" leadership style as one very conducive of "good behavior" and good examples to be followed. According to the interviewee, "foreign" managers need to learn how things are done. So it appears that "doing things right" in company C is more reliant on elements of the surrounding national culture than on an corporate culture anchored in own values, policies and routines. "How we, as a human being, drives off our emotions and, especially in this area what you're discussing is, I think it's from my perspective anyways, 95% driving from the emotion, not from code of conduct". Corporate culture and national culture are thus deemed to be the same thing and unlikely to travel, meaning that different leadership styles, regardless of if they are conducive of good behavior or not, will set the management tone in different parts of the world.

Another interviewee in the product portfolio function, speaks about creating a strong brand: "We want to have a strong branding, of course. But that brand name shouldn't be just because it's a big company because we won't be as big as (the previous owner group) with 40,000-somewhat employees. But we want it to be that the brand is recognized as safe, nice to work with, backed by good relations, and good people, and all of those things. And in itself, we can do because we now own the brand (. . .). So we have a greater opportunity from that perspective. And of course, we, in a sense, own the distribution. We own all the channels towards the customer except we were represented by dealers. But at least

Sustainability **2021**, 13, 1986 14 of 20

we have a direct connection into the dealers to give us opportunities". However, at the same time, the brand needs to speak to customer needs, another important driver for the organization: "when we talk about sustainability, we more talk about our products and the way they are supposed to be used by the customers, I would say. More the customer focus". Recognizing that costumers in different parts of the world express different needs, also in terms of their sustainability requirements, the company ends up downplaying this ambition to meet current market needs.

3.4. How Internal Work Is Structured

All three organizations presented differences as to how social sustainability initiatives are driven and structured. Organizations A and C, however, had several aspects in common. Ecological sustainability and social sustainability in organizations A and C are treated as two separate aspects. Ecological sustainability, often referred to as "sustainability", is anchored at the tactical level of the organization, with functions such as purchasing, quality and compliance sharing aspects of ecological sustainability initiatives, but still using the same umbrella concept. These different functions collaborate with each other and provide support to product development. Social sustainability on the other hand, is not clearly assigned to any function, but different themes targeting different stakeholders (employees, suppliers, communities) are somewhat part of different functions at the tactical level of the organizations, such as human resources caring for compliance with collective agreements and employment regulations and health and safety making sure manufacturing processes do not create obstacles to employees' health. To some degree, purchasing functions also anchor some social sustainability aspects with respect to first tier suppliers. From the interviews conducted, we could not see any structured collaboration between these functions to define strategy, implementation or progress. At organization A we could see a growth in interest from product development professionals to work with sustainability more holistically and therefore improve their capabilities to work with social sustainability as well. Organizational structure and processes to facilitate bottom-up change and innovation were not a reality then.

Another interesting aspect that is common to organizations A and C is that both used to belong to corporate groups that had a stronger top-down steering style with sustainability strategy being defined at the top and trickled down to the operational level through policies, guidelines and checklists. Interviewees at organization A who experienced the previous management, share fond experiences of that time when they felt part of a bigger purpose. In the case of organization C, however, some interviewees mention the benefits of having clear guidelines and priorities set while others complained of a lack of trust from top management and limited freedom to innovate. One interviewee from company C describes being part of a group that had clear rules and requirements that had to be fulfilled internally and also in the relationship with suppliers (first tier). After becoming part of a new group, where no clearly defined guidelines were provided and a lot of freedom was given, several of these practices slowly ceased to exist according to the interviewee. "It was a listed company with a very strong focus on a lot of things. The way we do things was (. . .) clearly defined rules how to approach certain things. Now we are in a group that is more—it's not listed. It's a privately owned company. I think companies belonging to the new Group are more independent. There are no large group functions, really, telling everyone this is how we do it, blah, blah, blah. It doesn't really exist".

Organization B in contrast, includes social sustainability alongside ecological sustainability under the umbrella of sustainability. More precisely, it is anchored at the vice-presidency as Corporate Social Responsibility and Sustainability function, and in collaboration with the board of the organization and other counterparts at top-management level the strategic guidelines are defined. From top-management, functions in R&D, technology, marketing and after sales and human resources, receive the mandate to develop specific strategies, implement and monitor progress, reporting back to top management for consolidation and reporting. One interviewee talks about the vertical flow of information

Sustainability **2021**, 13, 1986 15 of 20

and support for decision-making: "There are no standards defined and such. We have in the policy some recommendations, or that there is guidance in the policy that our leaders and employees need to follow. But if there is risks, this is then escalated to council. And the council makes then a judgment, and it's always a subjective judgment. and this is made by specialists. And we have two important specialists in the council. One from the legal departments, anyone from the CSR department, and they will then recommend it to business owners how to act in this case."

Interviews showed that having one common framework that defined priority areas and high-level requirements creates greater awareness of societal social sustainability issues and also what those mean to specific functions of the organization. Apart from clear organizational structures, regular cross-function dialogue was identified as an important practice by interviewees, because it creates an environment of constant learning and striving for improvement. "Quite a lot of what we are doing here is based on the values of the company, but also the values of the employers here. And for example, in my leadership group, we have quite divided background, but we all share a quite positive approach about our core values environment, social care, these kind of discussions, we invite other people from other parts of the company and both internally and externally, locally and globally, to speak about what's happening in the society to speak about their environmental tasks, to speak about the challenge and I would say the world is facing forward."

4. Discussions and Conclusions

4.1. Moving Organizational Boundaries towards a Systems Perspective

The aim of this paper was to provide empirical evidence of how social sustainability is currently included in PDOs and what their motivations are to do so. In addition, we wanted to assess if these companies are in any way strategic in their social sustainability work. We were able to notice several differences between the three case companies' approach to social sustainability, in terms of how far they have advanced the integration of social sustainability practices throughout the organization, and in terms of the presence or absence of strategic thinking elements in the approach. A first key difference refers to the definition of the organization boundaries and their recognition of interdependencies with other social systems, be those other organizations, communities or society in general. In organization B the awareness of the complexity of social sustainability and the implications for them seem to be present in the whole management of the organization. Their lens goes beyond traditional organizational boundaries, meaning that targets and actions are being taken not only with an internal company focus but also upstream and downstream their value chain. Interestingly we found that there is a sense of civic duty or positive legacy towards people in all three organizations, although for company B it is much more pronounced through their values-driven leadership work. However, while this sense of responsibility for organization B embraces a wide range of interconnected social systems, for organizations A and C, who seem to have a more limited understanding of their social sustainability interdependencies within the value chain, their sense of responsibility to address social problems appears to be more limited, being directed only towards their own employees. The HR function putting the responsibility on "foreign" leadership provides a good example of a more limited understanding of system interdependencies. HR departments in Sweden-based organizations usually have the basic social sustainability work already laid out for them, due the country's robust social welfare system and thoroughly unionized workforce. Creating such conditions in countries where similar foundations are absent require a much more proactive attitude from HR departments and possibly Sweden-based HR departments do not see this as being in their scope or responsibility, since in their home country it is not.

Previous models of Corporate Social Responsibility (CSR) and Corporate Sustainability also identify the setting of organizational boundaries as a central differentiator of corporate strategies. Aggerholm and Trapp [42] outline three generations of CSR approaches where the first two generations are focused on the functioning of the organization itself and slowly include closely connected external stakeholders. Dyllick and Muff [43] talk about sustainability strategies 1.0 and 2.0 mainly taking an "inside-out" organizational

Sustainability **2021**, 13, 1986 16 of 20

perspective and only at 3.0 does it take an outside—in perspective. This means that it is only when companies begin seeing their contribution to (un)sustainability from a larger systems perspective that they can effectively contribute to addressing global challenges and create value for the common good. Dyllick and Rost [44] also posit that organizations can only claim that their products contribute to sustainability when they begin to systematically address societal challenges in a holistic way (both social and ecological aspects).

Expanding the scope to other systems creates the possibility for a wider range of stakeholders to be identified and also influence the social sustainability agenda and the breadth of topics considered relevant for the organization to tackle. While organizations A and C take a more reactive approach to stakeholder demands, such as employee collective agreements or industry requirements, organization B seems to actively engage a wider range of value chain stakeholders in the development of solutions to common challenges. Bolton et al. [45] have raised concerns over the stakeholder paradigm, such as that there is usually a focus on particular set of stakeholders, that it neglects the unequal possibility stakeholder groups might have to make their voice heard, and that it leads to taking a fire-fighting approach steered by maintaining corporate reputation, rather than a holistic and strategic approach.

Building on the above, organization B showed a broader coverage of social sustainability issues within their work whereas A and C stayed limited to internal issues such as health and safety of their own employees and some aspects of their products, particularly during the use phase. The broader awareness and perspective also allowed organization B to be more proactive and strategic in their social sustainability work while the other two companies were guided by existing local practices and regulations. This shows that it is hard to decouple how organizations identify (or not) themselves as part of social systems and their ability to develop social sustainability strategies that can contribute to addressing social challenges more widely.

A more systems-oriented framework, such as the FSSD, can support organizations in dealing with the stakeholder paradigm predicament as it asks organizations to be positioned as nested within socio—ecological systems and identify the interdependencies that create the conditions for their own long-term survival. In doing so, a wider lens is created for organizations to include stakeholders that would otherwise be missed in the process. The FSSD also provides boundary conditions for success (principles) based on this understanding of social systems functioning so that the organization can better identify root causes of social problems and develop solutions that have long-lasting impact. Furthermore, the use of the principles can help address Bolton et al.'s [45] critique of firefighting and lack of inclusiveness as organizations become better equipped to anticipate problems and be proactive in their social sustainability work.

For example, the first principle of social sustainability of the FSSD states that "in a sustainable society, people are not subject to structural obstacles to health" [37]. This means that people are not exposed to social conditions that systematically undermine their possibilities to avoid injury and illness; physically, mentally or emotionally. The principle does not stipulate minimum wage, maximum work hours, or ergonomic requirements, because it should be used by any social system, any organization, and not only today, but also in the future. International standards and best practice examples can support organizations in creating goals and a set of performance indicators, but due to the dynamic property of social systems, without clear guiding principles it becomes difficult to know if actions are taking organizations in the right direction. Artificial intelligence or increased digitalization of workplaces are good examples. Who could have predicted ten years ago the level of automation in our industries or the level of virtuality of social relations? New ways of creating healthy work environments need to be developed and new indicators must follow [46]. Our interviews indicate that while organizations A and C show a high skill level as regards working with health and safety issues, that skill was not always used to help address health and safety issues in the supply chain and sometimes among subsidiaries. Although in organization B we found that there is evidence of systematic

Sustainability **2021**, 13, 1986 17 of 20

work to prevent injury and illness of value chain actors in addition to own employees, we could not verify a clear high-level framework to help define and guide present and future actions. Top management is tuned into the latest trends and stakeholders demands, but not necessarily taking a precautionary principle approach to planning.

4.2. Implications for Organizational Structure and Process

In terms of setting targets related to social sustainability at organizational and project levels and defining related product requirements, organization B defined those at the strategic management level in the organization and ensured that lower levels of the organization were able to formulate specific targets accordingly. The other two did not have such a centralized decision-making forum and different parts of the organization decided on social sustainability issues they deemed important.

Based on the results of this study, it seems that better understanding of the complexity of social sustainability and a broader perspective of social sustainability interdependencies goes hand-in-hand with a way of organizing that overcomes traditional hierarchies and allows for more collaborative and strategic work in this area. At the same time, strong commitment from top level management creates opportunities for different roles and functions to play a part in whatever changes are deemed necessary. The literature on eco-design as well as CSR points to the importance of organizational structures to ensure sustainability integration. Brones and de Carvalho [10] state that macro, meso and micro levels of an organization should be integrated with top-down and bottom-up flows promoting vertical integration, while a transversal integration should occur through change and people management based on the organization's culture. Asif et al. [47] describe full CSR integration as "elements of organizational strategy cascade down to all levels and, thus, create a fit among organizational objectives, targets, and processes" and that "coordination among the departments is key to ensuring smooth processes and a coherent pursuit of corporate objectives to efficiently use resources and competencies" at the operational level is key. In organizations A and C we could identify such structures in place to deal with sustainability goals, but those most often referred to ecological aspects, possibly due to a narrow definition of sustainability. In organization B, who worked with a more holistic definition of sustainability that includes social aspects alongside ecological ones, organizational structures ended up serving social objectives equally well, allowing strategic social sustainability work to happen.

4.3. Implications for Sustainable Product Development

Last but not least, our research sheds light into aspects specific to product design and development. Sustainable product development, according to Hallstedt and Isaksson [48], is when a strategic sustainability perspective is integrated and implemented into the early phases of the product innovation process, that is, in the design stages, including lifecycle thinking. Schulte and Hallstedt [49] further detail that, for this to happen, changes in, processes and support tools are required across all levels of decision-making; that is, the strategic, tactical, and operational levels of the organization. Our case study brings evidence that currently very few social sustainability impacts connected to a product's lifecycle stages are managed by product developers within design related decisions, besides design of manufacturing processes and some user related activities. In all three organizations, health and safety issues connected to manufacturing activities featured high in the product developers' agenda, similarly to user safety. Social sustainability impacts connected to raw material extraction, transportation or end-of-life activities, when taken into consideration, appear to be dealt with by other functions outside design activities, such as purchasing or marketing although in rare occasions in dialogue with product developers. Product developers, however, have an opportunity to influence the social sustainability performance of a product by focusing their design efforts on the use phase, not in meeting expressed customer needs, but focusing on needs from a societal perspective. According to Bjørn et al. [50], changes in how things are produced must be enhanced by changes in what is

Sustainability **2021**, 13, 1986 18 of 20

being produced. Dyllick and Rost [44] talk about an enhancement of the product life cycle through positive handprints, contributing to public value creation as a result from the primary product purpose. Because product development organizations should be asking how their products help meet the needs (as opposed to wants) of current and future generations [50], design activities also need to be supported by higher levels of decision-making and aligned business models.

Meuer et al. [51] identify three essential attributes that differentiate shades of corporate sustainability: (1) the level of ambition, (2) the level of integration and (3) the specificity of sustainable development. A joint appraisal of these attributes helps determine if firms are in fact contributing to sustainable development or only creating a marketing piece. In this study we have explored all three essential attributes. The authors stress, however, the importance of being clear and specific with what is included in a firm's definition of sustainability. Moreover, the authors acknowledge that sustainability progress and maturity do not consist of a single linear journey, but multiple and uneven ones, as different sustainability challenges may bring different managerial implications. The relevance of our research speaks directly to this. Because social sustainability has been overlooked for so long in PD research, there is a need to build the foundations in this area and develop descriptive research that contributes to a greater understanding of how companies currently approach social sustainability and what challenges they might face in further integrating it in organizational practices. Previous theories within SPD and related areas might transfer well to the challenges that social sustainability pose to PDOs, but this should not be taken for granted as there is a possibility for significant differences and therefore needs to be done critically assessing implications to practice or new theory development.

Finally, we would like to express that exploring social sustainability with our interviewees has led us to confirm some of our assumptions but also to challenge others, starting with the concept of social sustainability itself. We came to realize that often times it needed explaining, even to functions that we expected to be well versed in it. Other times, the depth of understanding of its foundations and implications was high, even if it was not wrapped by a unifying concept. Some interviews flowed easily, and respondents felt comfortable in discussing the topic. Others struggled, were puzzled by the questions, and felt they had nothing to contribute with. The majority of those we interacted with, however, expressed curiosity and a wish to learn more in order to have their work contribute to a better world.

5. Limitations

Despite the above-mentioned contributions, this study has some limitations that need to be acknowledged. Data were collected from only three organizations, all of them located in Sweden. Although their sectors slightly differed, the study still focused on an industrial business area that presents several similarities in terms value chain complexity and business to business activities. This was an exploratory study based on a case study design, which naturally excludes a statistical generalizability of our findings. Such generalizability should be tested in future work using a greater number of companies from different sectors and also operating in different national contexts [32]. In addition, only a limited number of interviews could be conducted at the case companies, especially at organizations B and C, which makes the phenomenon in this study, even if common for this type of research, closely connected to our interviewees' perceptions. Since the selected participants are key organizational members that should enact social sustainability strategies, we argue that their views constitute relevant evidence of the different social sustainability lenses present at the case organizations. Finally, we acknowledge the time delay between conducting interviews and the writing of this article, which could mean that the participating companies have advanced in their social sustainability integration work. The final version of the article has, however, been approved by the companies.

Sustainability **2021**, 13, 1986 19 of 20

Author Contributions: Conceptualization, P.L.M. and M.M.; methodology, P.L.M.; validation, P.L.M.; formal analysis, P.L.M.; investigation, P.L.M.; data curation, P.L.M.; writing—original draft preparation, P.L.M. and M.M.; writing—review and editing, P.L.M. and M.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research and the APC were funded by the Knowledge Foundation and the FUTURA Foundation, both in Sweden.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Not applicable.

Acknowledgments: Financial support from the Knowledge Foundation and the FUTURA Foundation, both in Sweden, is gratefully acknowledged. Sincere thanks are given to the industrial partners and interviewees who participated in the study, and all the support and inputs received from Sophie Hallstedt and Göran Broman.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

References

- 1. Tischner, U.; Charter, M. Sustainable Product Design. In *Sustainable Solutions: Developing Products and Services for the Future*; Routledge: London, UK, 2017.
- 2. United Nations. *Transforming Our World: The 2030 Agenda for Sustainable Development;* Division for Sustainable Development Goals: New York, NY, USA, 2015.
- Tan, A.R.; McAloone, T.C.; Myrup Andreasen, M. What happens to integrated product development models with product/servicesystem approaches? In Proceedings of the IPD 2006: Proceedings of the 6th Workshop on Integrated Product Development, Magdeburg, Germany, 18–20 September 2006.
- 4. Sarkis, J. Manufacturing's role in corporate environmental sustainability-Concerns for the new millennium. *Int. J. Oper. Prod. Manag.* **2001**, 21, 666–686. [CrossRef]
- 5. Hauschild, M.; Jeswiet, J.; Alting, L. From Life Cycle Assessment to Sustainable Production: Status and Perspectives. *CIRP Ann.* **2005**, *54*, 1–21. [CrossRef]
- 6. Simons, L.; Slob, A.; Holswilder, H.; Tukker, A. The Fourth Generation: New Strategies Call for New Eco-Indicators. *Environ. Qual. Manag.* **2001**, *11*, 51–61. [CrossRef]
- 7. Karlsson, R.; Luttropp, C. EcoDesign: What's happening? An overview of the subject area of EcoDesign and of the papers in this special issue. *J. Clean. Prod.* **2006**, *14*, 1291–1298. [CrossRef]
- 8. Nochur, K.S. Design for Environment: Creating Eco-Efficient Products and Processes; Fiksel, J., Ed.; McGraw Hill: New York, NY, USA, 1996.
- 9. International Organization for Standardization. ISO 14040: Environmental Management–Life Cycle Assessment—Principles and Framework; International Organization for Standardization: Geneva, Switzerland, 2006.
- 10. Brones, F.; De Carvalho, M.M. From 50 to 1: Integrating literature toward a systemic ecodesign model. *J. Clean. Prod.* **2015**, *96*, 44–57. [CrossRef]
- 11. Faludi, J.; Hoffenson, S.; Kwok, S.Y.; Saidani, M.; Hallstedt, S.I.; Telenko, C.; Martinez, V. A Research Roadmap for Sustainable Design Methods and Tools. *Sustainability* **2020**, *12*, 8174. [CrossRef]
- 12. Hutchins, M.J.; Sutherland, J.W. An exploration of measures of social sustainability and their application to supply chain decisions. *J. Clean. Prod.* **2008**, *16*, 1688–1698. [CrossRef]
- 13. McMahon, M.; Bhamra, T. 'Design Beyond Borders': International collaborative projects as a mechanism to integrate social sustainability into student design practice. *J. Clean. Prod.* **2012**, *23*, 86–95. [CrossRef]
- 14. Gmelin, H.; Seuring, S. Determinants of a sustainable new product development. J. Clean. Prod. 2014, 69, 1–9. [CrossRef]
- 15. Littig, B.; Griessler, E. Social sustainability: A catchword between political pragmatism and social theory. *Int. J. Sustain. Dev.* **2005**, *8*, 65. [CrossRef]
- 16. Partridge, E. Social Sustainability': A Useful Theoretical Framework; Conference Papers; APSA: Washington, DC, USA, 2005.
- 17. Kunz, J. Social Sustainability and Community Involvement in Urban Planning; University of Tampere: Tampere, Finland, 2006; p. 118.
- 18. Cuthill, M. Strengthening the 'social' in sustainable development: Developing a conceptual framework for social sustainability in a rapid urban growth region in Australia. *Sustain. Dev.* **2010**, *18*, 362–373. [CrossRef]
- 19. Dempsey, N.; Bramley, G.; Power, S.; Brown, C. The social dimension of sustainable development: Defining urban social sustainability. *Sustain. Dev.* **2009**, *19*, 289–300. [CrossRef]
- 20. Vallance, S.; Perkins, H.C.; Dixon, J.E. What is social sustainability? A clarification of concepts. *Geoforum* **2011**, 42, 342–348. [CrossRef]

Sustainability **2021**, 13, 1986 20 of 20

 Weingaertner, C.; Moberg, A. Exploring Social Sustainability: Learning from Perspectives on Urban Development and Companies and Products. Sustain. Dev. 2011, 22, 122–133. [CrossRef]

- 22. McMahon, M.; Bhamra, T. Social Sustainability in Design: Moving the Discussions Forward. Des. J. 2015, 18, 367–391. [CrossRef]
- 23. Findeli, A. Sustainable Design: A Critique of the Current Tripolar Model. Des. J. 2008, 11, 301–322. [CrossRef]
- 24. Jørgensen, A.; Le Bocq, A.; Nazarkina, L.; Hauschild, M. Methodologies for social life cycle assessment. *Int. J. Life Cycle Assess.* **2007**, *13*, 96–103. [CrossRef]
- 25. Wu, R.; Yang, D.; Chen, J. Social Life Cycle Assessment Revisited. Sustainability 2014, 6, 4200–4226. [CrossRef]
- 26. Baumann, H.; Boons, F.; Bragd, A. Mapping the green product development field: Engineering, policy and business perspectives. *J. Clean. Prod.* **2002**, *10*, 409–425. [CrossRef]
- 27. Boks, C. The soft side of ecodesign. J. Clean. Prod. 2006, 14, 1346–1356. [CrossRef]
- 28. Schulte, J.; Hallstedt, S.I. Self-Assessment Method for Sustainability Implementation in Product Innovation. *Sustainability* **2018**, 10, 4336. [CrossRef]
- 29. Watz, M.; Hallstedt, S.I.; Watze, M. Profile model for management of sustainability integration in engineering design requirements. J. Clean. Prod. 2020, 247, 119155. [CrossRef]
- 30. Wang, H.; Gibson, C.; Zander, U. Editors' Comments: Is Research on Corporate Social Responsibility Undertheorized? *Acad. Manag. Rev.* **2020**, 45, 1–6. [CrossRef]
- 31. Robson, C. Real World Research; Wiley-Blackwell: Hoboken, NJ, USA, 2002; ISBN 978-1-4051-8240-9.
- 32. Yin, R.K. Case Study Research: Design and Methods, 5th ed.; SAGE: London, UK, 2014; ISBN 978-1-4522-4256-9.
- 33. Johnson, K.E.; Stake, R.E. The Art of Case Study Research. Mod. Lang. J. 1996, 80, 556. [CrossRef]
- 34. Braun, V.; Clarke, V. Using thematic analysis in psychology. Qual. Res. Psychol. 2006, 3, 77–101. [CrossRef]
- 35. Broman, G.I.; Robèrt, K.-H. A framework for strategic sustainable development. J. Clean. Prod. 2017, 140, 17–31. [CrossRef]
- 36. Missimer, M.; Robèrt, K.-H.; Broman, G. A strategic approach to social sustainability—Part 1: Exploring the social system. *J. Clean. Prod.* **2017**, *140*, 32–41. [CrossRef]
- 37. Missimer, M.; Robèrt, K.-H.; Broman, G. A strategic approach to social sustainability—Part 2: A principle-based definition. *J. Clean. Prod.* **2017**, 140, 42–52. [CrossRef]
- 38. Robèrt, K.-H. Tools and concepts for sustainable development, how do they relate to a general framework for sustainable development, and to each other? *J. Clean. Prod.* **2000**, *8*, 243–254. [CrossRef]
- 39. Robèrt, K.-H.; Schmidt-Bleek, B.; De Larderel, J.A.; Basile, G.; Jansen, J.; Kuehr, R.; Thomas, P.P.; Suzuki, M.; Hawken, P.; Wackernagel, M. Strategic sustainable development—Selection, design and synergies of applied tools. *J. Clean. Prod.* **2002**, *10*, 197–214. [CrossRef]
- 40. Dreborg, K.H. Essence of backcasting. Futures 1996, 28, 813–828. [CrossRef]
- 41. Missimer, M.; Robèrt, K.-H.; Broman, G. A systems perspective on ISO 26000. In Proceedings of the 2nd International Symposium "Systems Thinking for a Sustainable Aconomy. Advancements in Economic and Managerial Theory and Practice, Roma, Italy, 23–24 January 2014.
- 42. Aggerholm, H.K.; Trapp, N.L. Three tiers of CSR: An instructive means of understanding and guiding contemporary company approaches to CSR? *Bus. Ethic A Eur. Rev.* **2014**, 23, 235–247. [CrossRef]
- 43. Dyllick, T.; Muff, K. Clarifying the Meaning of Sustainable Business. Organ. Environ. 2016, 29, 156–174. [CrossRef]
- 44. Dyllick, T.; Rost, Z. Towards true product sustainability. J. Clean. Prod. 2017, 162, 346–360. [CrossRef]
- 45. Bolton, S.C.; Kim, R.C.-H.; O'Gorman, K.D. Corporate Social Responsibility as a Dynamic Internal Organizational Process: A Case Study. *J. Bus. Ethic* **2011**, *101*, *61*–74. [CrossRef]
- 46. Isaksson, O.; Eckert, C. *Product Development 2040: Technologies are Just as Good as the Designer'S Ability to Integrate Them*; The Design Society: Glasgow, UK, 2020; Volume 56, pp. 1–56. [CrossRef]
- 47. Asif, M.; Searcy, C.; Zutshi, A.; Fisscher, O.A. An integrated management systems approach to corporate social responsibility. *J. Clean. Prod.* **2013**, *56*, 7–17. [CrossRef]
- 48. Hallstedt, S.I.; Isaksson, O. Material criticality assessment in early phases of sustainable product development. *J. Clean. Prod.* **2017**, *161*, 40–52. [CrossRef]
- 49. Schulte, J.; Hallstedt, S.I. Company Risk Management in Light of the Sustainability Transition. *Sustainability* **2018**, *10*, 4137. [CrossRef]
- 50. Bjørn, A.; Bey, N.; Georg, S.; Røpke, I.; Hauschild, M.Z. Is Earth recognized as a finite system in corporate responsibility reporting? *J. Clean. Prod.* **2017**, *163*, 106–117. [CrossRef]
- 51. Meuer, J.; Koelbel, J.; Hoffmann, V.H. On the Nature of Corporate Sustainability. Organ. Environ. 2019, 33, 319–341. [CrossRef]